Author’s response to reviews

Title: Inherited thrombotic thrombocytopenic purpura mimicking Immune Thrombocytopenic Purpura: a case report during pregnancy

Authors:

VALTER ROMÃO DE SOUZA JUNIOR (jr_walter@hotmail.co.uk)

Ana Beatriz Cavalcante de Oliveira (anabeatrizoliveira22@hotmail.com)

Ana Maria Vanderlei (manuhazin@yahoo.com)

Amanda Queiroz da Mota Silveira Aroucha (amanda.motasilveira@gmail.com)

Bruna Duarte (brunapduarte@gmail.com)

Aureli Nunes Machado (aurelinnm@gmail.com)

Lívia Netto Chaer (livianettochaer@gmail.com)

Cláudia Wanderley de Barros Correia (claudia.wanderley@uol.com.br)

Maria da Conceição de Barros Correia (cbarroscorreia@yahoo.com.br)

Manuela Freire Hazin Costa (manuhazin@yahoo.com.br)

Version: 2 Date: 27 Nov 2017

Author’s response to reviews:

Reviewer #2:

1) There are NOT still some typos and misused abbreviations, which need to rechecked and corrected.

2) In figures 1&3, the WE called orange arrows'

3) The abstract submitted to the system WAS CHANGED.
Reviewer #3: Dear Reviewer,

1) WE Corrected PTT to TTP (page 3 line 43).

2) we Corrected platelet units to $x10^9$/L (page 3 lines 22,37)

3) we Corrected RBC count units to $\times 10^{12}$/L, WBC units should be in $x10^9$/L, platelets units should be $x10^9$/L , fibrinogen in g/L(page 8 table 1).